Author's response to reviews

Title: An automated, broad-based, near real-time public health surveillance system using presentations to hospital Emergency Departments in New South Wales, Australia

Authors:

David J Muscatello (dmusc@doh.health.nsw.gov.au)
Tim Churches (tchur@doh.health.nsw.gov.au)
Jill Kaldor (jkald@doh.health.nsw.gov.au)
Wei Zheng (wzhen@doh.health.nsw.gov.au)
Clayton Chiu (cchiu@doh.health.nsw.gov.au)
Patricia Correll (patriciac@woolcock.org.au)
Louisa Jorm (ljorm@doh.health.nsw.gov.au)

Version: 6 Date: 22 December 2005

Author's response to reviews:

All the the minor formatting changes requested by the editors have been made: 1) URLs as references, not in text; 2) Removal of "insert figure here" text; 3) Standardised text for competing interests section; 4) Reposition references before figure labels and tables; 5) Remove shading from tables.